Author's response to reviews

Title: Timing of operation for poor-grade aneurysmal subarachnoid hemorrhage: study protocol for a randomized controlled trial

Authors:

Qiao Zhang (26663782@qq.com)
Lu Ma (99353464@qq.com)
Yi Liu (5661322@qq.com)
Min He (hemiao98@163.com)
Hong Sun (sunhong1972hope@yahoo.com.cn)
Xiang Wang (4589235@qq.com)
Yuan Fang (18499320@qq.com)
Xu-hui Hui (huixuhui@126.com)
Chao You (doctorcastor@gmail.com)

Version: 2 Date: 2 July 2013

Author's response to reviews: see over
Dear editor:

Thank you for your constructive comments, we will give a point-by-point response to you according to your requirements.

1. Therefore, could you please clarify whether you have ethical approval and external funding for your study, and if so, could you send evidence of your approvals to the address below?

   Thank you for reminding. Yes, we have ethical approval and external funding for our study. We will send evidence of approval to you in Figure section of Upload files.

2. Could you also provide a list of any manuscripts that are under submission with other journals based on this study protocol?

   Null

3. We advise you to seek the assistance of a fluent English speaking colleague, or to have a professional editing service correct your language. Please ensure that particular attention is paid to the abstract.

   We have send our manuscript MS: 2518706599437178 to Edanz (www.edanzediting.com/bmc1) for improving quality of written English. And we will submit the revised version to you in Upload files.

4. Please include the name of the group within the list of authors, by adding the name to the 'first name' box; and the bracketed abbreviation to the 'last name' box in the submission system. Please remove the names of the members of the group from the title page and place them in the Acknowledgements, as they are to be acknowledged as members of the study group. The groups' contribution as a whole can be given in the authors contributions section.

   According to your opinion, we have revised our manuscript. You will see the updated versions in Upload files.